%%    TEMPLATE for articles submitted to the full-tex econf proceedings
%%     
%%
%%     Please do not remove lines commented out with %+
%%           these are for the editors' use.
%%
%%     Questions?  Send email to :  mpeskin@slac.stanford.edu
%%   

\documentclass[12pt]{article}
\usepackage{epsfig}
\usepackage{upgreek}
\usepackage{hyperref} 
%\usepackage{caption}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%  Do not change these:
\textwidth=6.0in  \textheight=8.25in

%%  Adjust these for your printer:
\leftmargin=-0.3in   \topmargin=-0.20in


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  personal abbreviations and macros
%    the following package contains macros used in this document:
\input econfmacros.tex
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  To include an item in the INDEX of the conference volume,
%           flag it with    \index{<item name>}
%  The use of this macro is illustrated in the text.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\Title#1{\begin{center} {\Large {\bf #1} } \end{center}}

\begin{document}
\begin{thebibliography}{99}
%\small
%%
%%  bibliographic items can be constructed using the LaTeX format in SPIRES:
%%    see    http://www.slac.stanford.edu/spires/hep/latex.html
%%  SPIRES will also supply the CITATION line information; please include it.
%%
\bibitem{DetPaper}
LHCb Collaboration, A. A. Alves Jr. {\it et al.}, 
{\it The LHCb detector at the LHC},
JINST {\bf 3} (2008) S08005.
\bibitem{MSTW08}
A.~D.~Martin, W.~J.~Stirling, R.~S.~Thorne and G.~Watt,
{\it Parton distributions for the LHC},
Eur.\ Phys.\ J.\ C {\bf 63} (2009) 189,
arXiv:0901.0002 [hep-ph].
%%CITATION = ARXIV:0901.0002;%%
%\bibitem{PDF4LHC}
%M.~Botje {\it et al.},
%{\it The PDF4LHC Working Group Interim Recommendations,}
%arXiv:1101.0538 [hep-ph].
%%CITATION = ARXIV:1101.0538;%%
%M. Botje {\it et al.}, The PDF4LHC Working Group Interim Recommendations, arXiv:1101.0538v1 [hep-ph].
\bibitem{ThorneLHCb}
R.~S.~Thorne, A.~D.~Martin, W.~J.~Stirling and G.~Watt,
{\it Parton Distributions and QCD at LHCb},
arXiv:0808.1847 [hep-ph].
%%CITATION = ARXIV:0808.1847;%%
\bibitem{MuMuPaper}
LHCb Collaboration, R. Aaij {\it et al.},
{\it Inclusive $W$ and $Z$ production in the forward region at $\sqrt{s} = 7$ TeV},
JHEP {\bf 1206} (2012) 058,
arXiv:1204.1620 [hep-ex].
%%CITATION = ARXIV:1204.1620;%%
\bibitem{eeConf}
LHCb Collaboration, R. Aaij {\it et al.},
{\it Measurement of the cross-section for $\mathrm{Z}^0\nolinebreak\rightarrow\nolinebreak\mathrm{e^+\nolinebreak e^-}$ production in pp collisions at $\sqrt s = 7$~TeV},
LHCb-CONF-2012-011.
\bibitem{tautauConf}
LHCb Collaboration, R. Aaij {\it et al.},
{\it $\mathrm{Z}$ cross-section measurement at $\sqrt s = 7$~TeV using the channel $\mathrm{Z}\nolinebreak \rightarrow \nolinebreak\uptau \uptau$},
LHCb-CONF-2011-041.
\bibitem{ZJetConf}
LHCb Collaboration, R. Aaij {\it et al.},
{\it Measurement of jet production in $\mathrm{Z}^0/\upgamma^*\nolinebreak\rightarrow\nolinebreak\upmu^+\nolinebreak\upmu^-$ events at LHCb in $\sqrt s = 7$~TeV $\mathrm{pp}$ collisions},
LHCb-CONF-2012-016.
\bibitem{antikt}
%@article{lh\bibitem{Cacciari:2008gp}
M.~Cacciari, G.~P.~Salam and G.~Soyez,
{\it The Anti-k(t) jet clustering algorithm},
JHEP {\bf 0804} (2008) 063,
arXiv:0802.1189 [hep-ph].
\bibitem{FEWZ}
R.~Gavin, Y.~Li, F.~Petriello and S.~Quackenbush,
{\it FEWZ 2.0: A code for hadronic Z production at next-to-next-to-leading order},
Comput.\ Phys.\ Commun.\  {\bf 182} (2011) 2388,
arXiv:1011.3540 [hep-ph].
%%CITATION = ARXIV:1011.3540;%%
\bibitem{DY}
LHCb Collaboration, R. Aaij {\it et al.},
{\it Inclusive low mass Drell-Yan production in the forward region at $\sqrt s = 7$~TeV},
LHCb-CONF-2012-013.

  %%CITATION = ARXIV:0802.1189;%%cb_detector,
%      author         = "Alves~Jr., A. A. and others",
%      title          = "{The \lhcb detector at the LHC}",
%      collaboration  = "LHCb Collaboration",
%      journal        = "JINST",
%      volume         = "3",
%      pages          = "S08005",
%      doi            = "10.1088/1748-0221/3/08/S08005",
%      year           = "2008",
%}
%%CITATION = PWASA,13,1564;%%
%\bibitem{Mesmer}
%F. A. Mesmer, Proc. Wien. Acad. Sci. {\bf 13}, 1564, 1593 (1762).
%%CITATION = PWASA,13,1564;%%

%\bibitem{diCenzo}
%A. L. di Cenzo, Trans. Acad. Ducal.  Milan., {\bf 23}, 2647 (1771).
%%CITATION = TAADD,23,2647;%%

%\bibitem{Muller}
%For an exhaustive review of the Prussian literature, see A. D. M\"uller,
%in Workshop on Action at a Distance, F. Eisenschmidt, ed. (Springer,
%Berlin, 1774).

%\bibitem{daPonte}
%L. da Ponte, Trans. N. Y. Acad. Sci., {\bf 3}, 27 (1795).
%%CITATION = TNYAA,3,27;%%


\end{thebibliography}



\Title{Physics with Electroweak Gauge Bosons at LHCb}

\bigskip\bigskip

%+\addtocontents{toc}{{\it W. Barter}}
%+\label{BarterStart}

\begin{raggedright}  

{\it William Barter\footnote{On behalf of the LHCb collaboration; contact: \href{mailto:barter@hep.phy.cam.ac.uk}{\texttt{barter@hep.phy.cam.ac.uk}}}\index{Barter, W.}\\
University of Cambridge,\\
Department of Physics,\\
Cavendish Laboratory,\\
J J Thomson Avenue,\\
Cambridge,\\
CB3 0HE, UK\\}
\bigskip\bigskip

\end{raggedright}



\section{Introduction}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include an .eps figure into your paper
%%
%%\begin{figure}[htb]
%%\begin{center}
%%\epsfig{file=magnet.eps,height=1.5in}
%%\caption{Plan of the magnet used in the Mesmeric studies.}
%%\label{fig:magnet}
%%\end{center}
%%\end{figure}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include a LaTeX table  into your paper
%%
%\begin{table}[b]
%\begin{center}
%\begin{tabular}{l|ccc}  
%Patient &  Initial level($\mu$g/cc) &  w. Magnet &  
%w. Magnet and Sound \\ \hline
% Guglielmo B.  &   0.12     &     0.10      &     0.001  \\
% Ferrando di N. &  0.15     &     0.11      &  $< 0.0005$ \\ \hline
%\end{tabular}
%\caption{Blood cyanide levels for the two patients.}
%\label{tab:blood}
%\end{center}
%\end{table}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The LHCb detector is a single arm spectrometer fully instrumented in the forward pseudorapidity ($\eta$) region $2.0<\eta<5.0$. It has been designed to search for new physics in the decays of B and D mesons in proton-proton
collisions, but it is also well suited for studies of QCD and Electroweak physics. The pseudorapidity coverage of LHCb overlaps with the general purpose detectors (GPDs), ATLAS and CMS, in the region $2.0<\eta<2.5$, but uniquely covers the region $2.5<\eta<5.0$, allowing complementary studies to the GPDs. More information about the LHCb detector can be found in~\cite{DetPaper}. This contribution presents an overview of the LHCb results on electroweak boson production at $\sqrt s = 7$~TeV.

Measurements of electroweak boson production are an important benchmark of Standard Model processes at the LHC, and allow constraints to be placed on PDFs~\cite{MSTW08}. In these measurements, LHCb provides unique information~\cite{ThorneLHCb}. As LHCb occupies the forward region, collisions require one parton from high Bjorken-$x$ and one from low $x$. PDFs at high $x$ are already well constrained by existing measurements. However, LHCb probes previously unexplored regions of phase space at low $x$ and a range of $Q^2$, from measurements of $\mathrm{W}$, $\mathrm{Z}$ and lower mass Drell Yan production. In this region, the PDF uncertainties on cross-sections are large, and LHCb plays an important role in providing new constraints. This and the kinematic region probed by LHCb are shown in Fig.~\ref{fig:xQ2}.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include an .eps figure into your paper
%%

\begin{figure}[htb]
  \begin{center}
  \epsfig{file=PDFprobes.eps,width=0.84\textwidth}
  %\vspace{-0.5cm}
  \caption[]{%\small 
{\emph {Left:}} The regions of $x$ and $Q^2$ probed by the different LHC experiments (with $\sqrt s = 14$~TeV) in relation to previous measurements.\\
{\emph {Right:}} The PDF uncertainty associated with the cross-section of various benchmark processes increases in the forward region ($\sqrt s = 14$~TeV is shown here). From~\cite{ThorneLHCb}.}
  \label{fig:xQ2}
  \end{center}
\end{figure}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Electroweak Boson Production at LHCb}
%\begin{figure}[htb]
%\begin{center}
%\epsfig{file=../figs/Zee/Fig4.eps,width=0.7\textwidth}
%\vspace{-0.25cm}
%\caption[]{%\small
%The cross-section for the $\mathrm{Z}\rightarrow\mathrm{ll}$ process measured in different channels. The data are given as bands (with the inner band showing the statistical uncertainty and the outer band showing the total uncertainty), with the theoretical predictions shown as points, for different PDF sets. All the LHCb results are consistent with each other, and with NNLO theoretical predictions. From~\cite{eeConf}.}
%\label{fig:AllZ}
%\end{center}
%\end{figure}
\begin{figure}
\begin{center}
\epsfig{file=Summary.eps,width=0.672\textwidth}
%\vspace{-0.5cm}
\caption[]{%\small 
LHCb measurements (bands) for the $\mathrm{W}\rightarrow\upmu\upnu$ and the $\mathrm{Z}\rightarrow\upmu\upmu$ cross-sections, and their ratios, are consistent with NNLO predictions (points). From~\cite{MuMuPaper}.}
\label{fig:WZRatio}
\end{center}
\end{figure}
LHCb has measured inclusive $\mathrm{Z}$ production in the dimuon~\cite{MuMuPaper}, dielectron~\cite{eeConf} and ditau~\cite{tautauConf} channels. These measurements are performed in the fiducial acceptance $60\nolinebreak < \nolinebreak M(\nolinebreak\mathrm{\ell\ell})\nolinebreak<\nolinebreak 120$~GeV, $2.0\nolinebreak<\nolinebreak\eta\nolinebreak(\nolinebreak\mathrm{\ell}\nolinebreak)\nolinebreak<\nolinebreak4.5$, $p_\mathrm{T}\nolinebreak(\mathrm{\ell})\nolinebreak >\nolinebreak 20$~GeV. In addition, inclusive $\mathrm{W}\rightarrow\upmu\upnu$ production has been measured~\cite{MuMuPaper}, in the fiducial acceptance $2.0\nolinebreak<\nolinebreak\eta\nolinebreak(\nolinebreak\mathrm{\nolinebreak\upmu}\nolinebreak)\nolinebreak<\nolinebreak4.5$, $p_\mathrm{T}(\mathrm{\upmu}) > 20$~GeV. The results in the muon channels are summarised in Fig.~\ref{fig:WZRatio}. The LHCb results are consistent with each other and with next-to-next-to-leading order (NNLO) predictions, and test the Standard Model up to an accuracy of 1.7\% (in the ratio of the $\mathrm{W}^+$ to $\mathrm{W}^-$ production cross-sections). The LHCb measurement of the $\mathrm{W}$ lepton charge asymmetry, shown in Fig.~\ref{fig:WA}, also probes PDF predictions as this distribution is particularly sensitive to the ratio of the up quark PDF to the down quark PDF~\cite{ThorneLHCb}. The LHCb acceptance covers an interesting region where this distribution changes sign ($\mathrm{W}^-$ production becomes favoured over $\mathrm{W}^+$ production). This feature is due to the $V-A$ structure of the weak interaction.
\begin{figure}[htb]
\begin{center}
\epsfig{file=CSWA20.eps,width=0.693\textwidth}
%\vspace{-0.5cm}
\caption[]{%\small
 The lepton charge asymmetry, $\frac{\sigma(\mathrm{W}^+)-\sigma(\mathrm{W}^-)}{\sigma(\mathrm{W}^+)+\sigma(\mathrm{W}^-)}$ as a function of $\eta(\upmu)$ measured at LHCb (shown as bands) is consistent with NNLO predictions (points). From~\cite{MuMuPaper}.}
\label{fig:WA}
\end{center}
\end{figure}

Jet production in $\mathrm{Z}\rightarrow\upmu\upmu$ events has also been measured~\cite{ZJetConf}. Jets are reconstructed using a particle flow approach, and clustered using the anti-$k_\mathrm{T}$ algorithm~\cite{antikt} with radius parameter $R=0.5$. The jet energies are corrected back to the hadron level. Jets are required to have $2.0\nolinebreak<\nolinebreak\eta\nolinebreak<\nolinebreak4.5$, $p_\mathrm{T}\nolinebreak>\nolinebreak10$~GeV and to be separated from both the decay muons of the $\mathrm{Z}$ in $\eta-\phi$ space by a distance $\Delta R > 0.4$. The fraction of jet events is found to be $0.229 \nolinebreak\pm\nolinebreak 0.006 \nolinebreak\pm\nolinebreak 0.009$, where the first uncertainty is statistical, and the second is systematic. The results are compared against a NLO prediction from FEWZ~\cite{FEWZ} using MSTW08 PDFs~\cite{MSTW08}, and found to be consistent.

LHCb has also measured low mass Drell-Yan production in the $\upmu\upmu$ channel~\cite{DY} in the fiducial acceptance $2.0<\eta(\upmu)<4.5$, $5<M(\upmu\upmu)<120$~GeV, $p(\upmu)\nolinebreak>\nolinebreak10$~GeV, and $p_\mathrm{T}(\upmu)\nolinebreak >\nolinebreak 3(15)$~GeV for $M(\upmu\upmu) > 5(40)$~GeV. NLO predictions agree reasonably well with the LHCb measurement.

\section{Conclusions}
Electroweak boson production has been studied extensively at LHCb. Measurements at LHCb are complementary to those from the GPDs. The results presented are consistent with Standard Model theory predictions, and can be used to provide new constraints on PDFs in a previously unexplored region of phase space.
%\bigskip




%\Discussion



%\speaker{NAME} Topic.

%\endDiscussion
 
\end{document}

